Author's response to reviews

Title: Methods for the thematic synthesis of qualitative research in systematic reviews

Authors:

James Thomas (j.thomas@ioe.ac.uk)
Angela Harden (a.harden@ioe.ac.uk)

Version: 4 Date: 2 July 2008

Author's response to reviews:

Dear Sir or Madam,

We have made the following changes to the manuscript:

-------------------------------------------

Major revisions (necessary)

Authors' contributions - Please state that both authors read and approved the final manuscript. This has been done.

Minor revisions (we can make these changes for you)

Country - Please remember to include the country in the affiliation details. 'UK' has been added to our affiliation details.

Figures - Please rename this section 'Figure legends'. This has been done.

Best wishes, James.